Author's response to reviews

Title: EphA4 is a prognostic factor in gastric cancer

Authors:

Kohji Miyazaki (k-miyazaki.srg2@tmd.ac.jp)
Mikito Inokuchi (m-inokuchi.srg2@tmd.ac.jp)
Yoko Takagi (takagi.srg2@tmd.ac.jp)
Keiji Kato (keiji-kato.srg2@tmd.ac.jp)
Kazuyuki Kojima (k-kojima.srg2@tmd.ac.jp)
Kenichi Sugihara (k-sugi.srg2@tmd.ac.jp)

Version: 2 Date: 14 December 2012

Author's response to reviews:

Dear, Journal editorial office of BioMed Central

As you mentioned,
(1) We added ethics statement and consent statement in the Methods section.
(2) We wrote competing interests at the end of this article.

Sincerely Yours,
Mikito Inokuchi
Department of surgical oncology, Tokyo Medical and Dental University
1-5-45 Bunkyo-ku, Yushima, Tokyo, 113-8519 Japan
TEL 81-3-5803-5261, FAX 81-3-5803-0139
m-inokuchi.srg2@tmd.ac.jp